Author's response to reviews

Title: The primary care amplification model: taking the best of primary care forward

Authors:

Claire L Jackson (c.jackson@uq.edu.au)
Deborah A Askew (d.askew@uq.edu.au)
Caroline Nicholson (caroline.nicholson@mater.org.au)
Peter M Brooks (p.brooks@uq.edu.au)

Version: 4 Date: 9 December 2008

Author's response to reviews:

Dear Editor

thank you for the oppourtunity to further refine our manuscript. We have made the following changes:

1) we have changed the title to sentence case
2) we have removed all underlining, coloured text and tracked changes
3) we have formatted the abbreviations as a paragraph rather than a list and changed the heading to Abbreviations
4) the weblink for the previous reference 8 is no longer active. We have removed this reference to avoid any confusion (the material that was included in this reference is included in the other references inserted at this point in the manuscript - ie. previous references 9 & 10)
5) we do not have any acknowledgements or additional files

We have also made a few, very minor editing changes in the manuscript.

regards

Deborah Askew
(on behalf of Claire Jackson, Caroline Nicholson and Peter Brooks)